

Article



An Assessment Tool to Identify the Financial Literacy Level of Financial Education Programs Participants' Executed by Ecuadorian Financial Institutions

Silvia Mariela Méndez-Prado^{1,*}, Vanessa Rodriguez¹, Kevin Peralta-Rizzo¹, Patricia Everaert², and Martin Valcke³

- ¹ Facultad de Ciencias Sociales y Humanísticas, Escuela Superior Politécnica del Litoral (ESPOL), Campus Gustavo Galindo Km. 30.5 Vía Perimetral, P.O. Box 09-01-5863, Guayaquil 090902, Ecuador
- ² Faculty of Economics and Business Administration, Ghent University, EB22, Sint-Pietersplein 7, 9000 Ghent, Belgium
- ³ Faculty of Psychology and Educational Sciences, Department of Educational Studies, Ghent University, Henri Dunantlaan 2, 9000 Ghent, Belgium
- * Correspondence: smendez@espol.edu.ec

Abstract: Due to the lack of a tool assessing the effectiveness of Financial Education Programs (FEP) imparted by the financial institutions (FI) in Ecuador by mandate, this research proposes and validates a scale to evaluate the Financial Literacy (FL) level of 314 participants (15–50 years old) after attending an FEP. The final 18-item FL scale, compounded with Financial Attitude, Financial Knowledge, and Financial Behavior dimensions, is reliable and robust according to the Exploratory and Confirmatory Factor Analysis. The average score for FL is 76 out of 100 points. The youngest scored the lowest, and the FI employees the highest. This contribution fulfills a needed FL scale to accurately measure the efforts of FI spreading the FE. It serves as a stepping stone for further research reconfirming its validity and applying the proposed FI's FL scale as a pre–post FL evaluation.

Keywords: financial literacy; financial education program; financial institutions; scale validation

1. Introduction

In recent years, the topic of financial literacy (FL) has had a growing interest, and it has become a major concern for governments on account of the economic and financial crisis of 2008, the growth of banking and insurance activity, and the pressure on consumer marketing [1]. Several studies around the world show the effect of FL on various relevant issues including savings and retirement planning [2–9]. Furthermore, FL has also been shown to be an important factor in the topics of stock market participation and investment decision-making [10–13]. A lack of FL can lead to poor saving and spending habits, the misuse of credit cards, and terrible investment choices, whereas the opposite entails that financially literate individuals are better equipped to make financial decisions, build a secure financial future, and reach their own life goals, improving economic stability [14].

FL is fundamental to making effective decisions across a range of financial contexts, improving financial well-being for individuals and society, and enabling participation in economic life. It is defined as the combination of financial awareness, knowledge, skills, attitudes, and behaviors. FL is not the same as financial education [15]. Financial education, on the other hand, refers to the process of improving the understanding of financial products, and the concepts and risks associated with achieving financial wellbeing. This process may include instruction and being provided with the information necessary for better decision-making [16].

There are efforts in place to respond to the issue of FL in most countries. Latin America and the Caribbean (LAC) call for Financial Education Programs (FEP) to counteract the FL issue, though there is scarce research focusing on this topic [17]. Financial education is the



Citation: Méndez-Prado, S.M.; Rodriguez, V.; Peralta-Rizzo, K.; Everaert, P.; Valcke, M. An Assessment Tool to Identify the Financial Literacy Level of Financial Education Programs Participants' Executed by Ecuadorian Financial Institutions. *Sustainability* **2023**, *15*, 996. https://doi.org/10.3390/ su15020996

Academic Editor: Adrian Castro-Lopez

Received: 27 November 2022 Revised: 30 December 2022 Accepted: 3 January 2023 Published: 5 January 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). main method of intervention whose expected result is to increase FL. Most studies aiming to impact FL levels design their interventions around financial education definitions [18–20]. These interventions show that FEPs have generated a positive impact. For example, the use of video vignettes on key financial decisions as an intervention to increase FL in Ecuador yielded positive results after a four-week long intervention [21]. In Brazil, similar results were reported from personal finance courses [22], and the effects of formal and business education [20].

In Mexico, the National Financial Education Committee was created in 2011 with a strategy in place to guide the country's efforts to promote financial well-being [23]. In Brazil, a National Strategy for Financial Education (ENEF) was created in 2010 to promote financial education actions free of charge and without commercial interest [24]. Similarly, in Chile, the ENEF was created in 2017 [25]. In Colombia, the Banking and Financial Institutions Association of Colombia (Asobancaria), establishes that FI under its regulations must provide adequate information or training regarding the financial products and services they offer, to develop skills and confidence that allow informed decision-making [26].

In the Ecuadorian case, a resolution instated by the Superintendency of Banks in 2013 mandated that all entities under its control must develop FEPs [27]. The FEP target group is composed of high school students, university students, and adults in general. An example of this is the 'Tus Finanzas' FEP which consists of a group of FIs conducting training to increase the financial capabilities of Ecuadorian society since 2014 [28]. However, there is no common framework to develop these programs. Until 2018, 12 out of 29 regulated institutions were providing both in-person and online FEPs, nine institutions offered only online programs, and one had only in situ classes [29].

Despite the efforts through FEPs, there is nonexistent control and no scale designed to evaluate these programs. This evaluation of the existing FEPs represents a fundamental first step for policymakers and interested parties to follow the necessary actions and prevent poor decision-making and improve financial well-being. In Ecuador, the scientific literature shows the design and validation of an FL scale, focused on Key Financial Decisions among university students [30], but nothing referring to the evaluation of the FI efforts in enhancing Ecuadorian FL.

Scales intended to measure FEPs are scarce in the LAC area [17]. More specifically, in Ecuador, there is a lack of customized scales to monitor the FEPs' impacts or the effects achieved. We are left with the uncertainty of not knowing if these programs are effective in their purpose of increasing FL levels. Being heterogeneous, an evaluation tool should be defined according to the indicated contents and the general guidelines of the FEPs' development.

Though there are scales that have been used worldwide and have been incredibly useful to make comparisons across different countries, including the OECD/INFE Toolkit for measuring FL and financial inclusion survey in 2018, the S&P FL survey in 2020, and The Test of FL survey in 2017 [31–33], among others, it is important to consider the heterogeneity of each country's framework [34]. FL is not a universal concept that can be measured across nations using the same questions, and therefore, questions should be adapted to a country's specificities [34]. Thus, to evaluate the FEPs of FIs in the specific context of Ecuador, we have decided to propose a different evaluation method.

In trying to fill this research gap, our study investigates the FL levels of the participants of FEPs imparted by FIs in Ecuador, improving the understanding of measurement scales in FL. The importance of this analysis lies in its potential for the improvement of FEPs, considering these new findings. In this study, we aim to make an exploration of the following research question: how financially literate are the participants of FEPs executed by Ecuadorian FIs? To answer this thoroughly, we define the following objectives:

First, we will propose the design of a scale to measure FL, using data from surveyed participants after receiving training from an FEP in Ecuador. Then, we will assess the validity and reliability of the proposed FL scale for FIs in Ecuador using Confirmatory Factor Analysis (CFA). Later, we will identify the FL levels of the participants after the intervention to characterize their performance in the Ecuadorian context and find potential determinants of the FL score using the sociodemographic data. Lastly, we will characterize the participants' preferences in the context of the banking products and services.

This paper is divided into five sections. Section 2 explains the conceptual basis of FL. Section 3 describes the sample, data collection process, and other details of the research design. Section 4 shows the statistical results. Section 5 discusses the findings. Lastly, Section 6 presents the main conclusions.

2. FL Conceptual Basis

2.1. FL and Its Dimensions

Throughout different studies, different interpretations of FL have been given. Notoriously, FL is understood as encompassing two main constructs or components: financial knowledge and the financial ability to properly apply this knowledge [35]. By recognizing that an individual may have financial knowledge but not have the ability to implement it, they would still be considered financially illiterate. To others, FL refers to the ability to process information about economic and financial concepts, for, without an understanding of these concepts, people will not be well-equipped in the financial decision-making processes [36]. The financial decisions faced by individuals in their day-to-day life include saving, investing, borrowing, and more. The OECD's [37] interpretation of FL regards three well-defined constructs: Financial Knowledge (FK), Financial Behavior (FB), and Financial Attitude (FA). Similarly, these same dimensions are considered when FL is referred to as "A combination of awareness, knowledge, skill, attitude, and behavior necessary to make sound financial decisions and ultimately achieve individual financial well-being" [38]. More recent studies are mostly based on previous definitions, especially the OECD, and encompass knowledge about financial topics, along with the ability to make sound financial decisions [39] such as the appropriate use of credit [40]. Other topics included in the FL definition pertain to the knowledge related to compound interest rates, risk management [41], budget management, money control, period planning, and the choice of financial products [42].

Based on these main definitions, there are three main dimensions of FL vastly regarded in the literature, with knowledge being the first one. For a person to be considered financially knowledgeable, one must have a solid understanding of key financial concepts, as well as numeracy skills whose application is essential in financial situations. These key concepts include simple and compound interest, risk and return, and inflation [38,43]. In a 2015 study, it was found that financial concepts may be more easily understood if one is proficient in math. Therefore, it is suggested that developing strong math knowledge and ability in young people may lead to better FL in adults in some countries [32]. Moreover, the correlation between FL and education is shown when the results of individuals with a university education or higher present better FK scores [44]. The link between FL and education should be clarified, since FK is a consequence of FEPs, and these are not the same [45].

FB is arguably the most important element of FL, as it has a direct impact on financial well-being, and is a determinant of FL [36,43]. Behaviors such as expenditure budgeting and securing a financial safety net are considered to indicate high levels of FL, while behaviors such as excessive credit use can indicate the contrary. An assessment of different studies on FL suggests a causal relationship that goes from financial knowledge to behavior [36]. They state that FL, which makes more of an emphasis on knowledge, influences financial decision-making. This has been studied using both experimental approaches, and instrumental variables [46–50].

Regarding the FA component, there is an important relationship between attitudes and the resulting FL. Attitudes encompass the knowledge, objective financial information, and the emotions associated which would result in an individual's observed behaviors [51]. The FA construct aims to measure how individuals self-evaluate their financial management abilities [52]. FA's relevance is highlighted by the following quote "If people have a rather

negative attitude towards saving for their future, for example, it is argued that they will be less inclined to undertake such behavior" [38]. This also suggests a relationship between behavior and attitude. The FA dimension considers a disjunctive between short-term gratification and long-term security in money matters [43], measuring attitudes toward planning for the future.

All previous research accentuates the importance of FK, in conjunction with the behavior and attitudes necessary to show improvement in their financial well-being, which is the goal [53–58].

These three components of FL (FK, FB, and FA) have become a common framework and have been used for multiple studies regarding the proposal of scales that measure FL.

This research proposes to test the three constructs (FK, FB, and FA) and their possible representative influences on FL to execute it in the Structural Equation Model (SEM), as it has been developed before [30,59].

2.2. Scales for Measuring FL

The importance of measuring FL has not gone unnoticed. Having such data provides policymakers with information regarding the specific needs and gaps of the population concerning FL. This allows them to provide financial education in the most effective way to the groups of people that need the most assistance [38].

Following their understanding of FL and the proposed dimensions (FK, FB, and FA), the OECD questionnaire proposes one of the most widely used and regarded FL scales. This report consists of a 21-item scale, where the FK score ranges between 0 and 7, the FB score ranges between 0 and 9, and the FA score ranges between 1 and 5. The overall FL score is calculated as a sum of the 21 items, meaning that in this scale, FB contributes the most to the final score [31]. The scale was validated by OECD/INFE experts and used to evaluate adults between the ages of 18 and 79 from 26 different countries. The results indicated an average score across all participating countries of just above 60% [15].

The Test of Financial Literacy is a 45-item test designed to measure the FK of high school students [33]. It was based on the National Standards for FL (Council for Economic Education 2013) which include the following topics: (1) earning an income; (2) buying goods and services; (3) saving; (4) using credit; (5) financial investing; and (6) protecting and insuring financial assets. Its main purpose was to help teachers to assess and improve the quality of personal finance instruction in high schools [60]. This test was validated and proven to be reliable (Cronbach's alphas = 0.87) and presented an average score of 43.48% for high school students. Though it was intended for this age group in the US, several variations have been used in different contexts [61,62].

Another relevant scale in the literature corresponds to the S&P Global FinLit Survey. This survey uses five questions to assess FL, which include fundamental financial topics necessary to make sound financial decisions (risk diversification, inflation, basic numeracy, and interest compounding) [32]. It is universal and applicable to every country, focused on knowledge and numeracy. If people know three out of the four topics, they are considered to be financially literate.

Some results that may be useful for comparisons in FL studies pertain to the OECD, according to whom the minimum average score for FL is 0.6 (60% of answers), approximately 0.71 for FK (71%), 0.6 for FB (or 60%), and 0.6 for FA (or 60%) [15]. Moreover, the S&P Global survey reports a score of around 60% for the most developed economies, while for emerging countries the score barely passed the 50% threshold [32].

The literary review of FL in LAC countries reports that about 50.77% of articles report measurement instruments or scales to evaluate FL in different contexts [17].

A study in LAC countries used a 25-item scale (Likert) to evaluate a sample of university students about FL topics that included savings, budget, credit management (cards and loans), investment, and financial terms. They considered the two dimensions of FK and FB. Their proposed model proved to be reliable (Cronbach's alphas = 0.681) [42]. Similarly, a study about credit card use and behavior evaluated young adults on the three dimensions, and validated their model obtaining good results (Cronbach's alpha > 0.69), determining the effect of this behavior on financial well-being [63].

Other studies in LAC countries present their scales, as original or variations from previous literature, however, not all of them consider the three dimensions. In Mexico, a 24-item FK scale has proven to be a reliable model (Cronbach's alphas = 0.860) [64]. Another Mexican study focuses on the FK dimension when analyzing students and their perception of financial tools [65]. Similarly, in Chile, a study about the determinants of personal loan total costs obtains a model (Cronbach's alphas > 0.7) through the lens of the knowledge dimension [66].

Several Brazilian studies on FL have designed and validated their scales. A study on FL and the gender gap also agrees with the three-dimensional model and proves the reliability of the scale (Cronbach's alphas = 0.6) [58]. Lastly, "Bolsa Família X" is a study that agrees with the three-dimensional model of FL and shows satisfactory psychometric results (Cronbach's alphas > 0.7) (23-item scale) [67].

Most recently, in Ecuador, a 44-item scale which focuses on Key Financial Decisions among young adults was designed and validated using the same dimensions previously discussed (FK, FB, and FA) [30]. They prove the validity and reliability of their scale with a sample from an Ecuadorian university. Their final model reflects good convergent validity (Cronbach's alpha = 0.855) among other positive statistic values. Their results show an average FL score of 65%. However, it focuses on key financial decisions, for example, retirement planning and the purchase of a house and car, while our proposal intends to evaluate a different set of topics based on the FIs' FEPs under the Superintendency of Banks of Ecuador (SB).

2.3. FEP in Ecuador

A 2020 FL and capabilities study in Ecuador found participants scored 12.2 out of 22 (55%), which was considered a high score, as it is within the Latin American average [68]. However, in a study with evidence from around the world, it was found that only 30% of the Ecuadorian sample was financially literate by correctly answering three out of five questions in the S&P FL Survey [69]. To understand the difference in results between these two studies, it is important to consider that these two are based on different methodologies and consider different criteria since the CAF study measures FL and compares it among Latin American countries while the second study mentioned presents evidence from around the whole world. The relatively deficient FL scores explain the reasoning behind the need for mandated FEPs. The SB issued a regulation asserting that all entities under its control must develop FEPs in favor of clients, collaborators, and the general public [27]. It is also specified that the FEP delivery system can be virtual or in situ to accommodate participants' needs. Additionally, it is stated that they must have evaluation mechanisms in place to certify the approval of the training. The training is free of cost and carried out through various modalities, including in-person financial education workshops and online courses [27], however, the methodologies vary depending on the institution imparting them [70–72]. A study from 2018 analyzed the 29 FIs' FEP performances regulated by the SB, where twelve private banks provide both online and in-person FEPs, nine only provide online FEPs, and one provides a solely in-person FEP [29]. Many of the FIs' FEPs have adapted to technological trends and now use different social media platforms such as Facebook, Instagram, Twitter, YouTube, radio shows, and webinars as channels to deliver their programs. This goes to show that the methodologies and channels through which people are educated vary according to the institution, as there is no specific format or learning delivery method they are required to follow.

In addition, the SB mandates that the main topics to be taught in the programs should include the structure and operation of controlled systems, banking products and services, savings and family budgeting, rights and obligations for the use of debit/credit cards, insurance products, and requirements to access social security benefits, among others specified in the scale design section. However, even when the topics comply with the

regulations, it is shown that usually the focus is on savings and family budgeting [29]. The duration of the FEP is also a heterogeneous factor within programs as there is no required number of hours specified in the resolution. Most FEPs do not indicate the hours taken to complete the program and only around 15% of them indicate the maximum duration of each module which ranges from 4 to 11 h [29]. The target audience follows SB guidelines; however, it has been found that 33.33% of institutions are focused on children and young adults, and 25.93%, are focused on clients and collaborators. Meanwhile, 93.10% of institutions are focused on women [29].

2.4. Sociodemographic Factors in FL

Sociodemographic factors are of relevance when studying FL scores [38,73,74]. Previous studies show an overwhelming difference between the FL levels of women and men, with women presenting lower scores. This trend is prevalent in important studies throughout the years and different places. In a study of FL around the world, it was shown that in most cases, women are less financially literate than men [73]. These results are consistent among countries according to international studies [38]. In Mexico, the three-question FL test encountered this same trend [40]. Others reiterate the same pattern [58,75,76]. This may be because men are more likely to relate money with power and social desirability [77], while women tend to have more passive and conservative attitudes [58].

Many researchers have found a strong correlation between FL and age, these relations being shaped by an inverted U [40,46,73]. This means that the levels of FL are lower in younger and older people, reaching a peak around middle age. While young people lack the experience that aids in acquiring FL, cognitive impairment might explain FL's decrease in older adults, which is why FL peaks in middle age [46]. As shown in a cross-sectional study where middle-aged adults made better decisions regarding the use of credit [46], the research determined that 53 years old was the age with minimized cost.

However, it is important to consider that high variability in FL has been found when it comes to young and old individuals [73]. For example, a study in Pakistan universities shows a significant FL difference between age groups among students, peaking at 26 in their sample [78].

Studies show that individuals with higher educational levels score better on FL tests [73], which is why FEPs are being applied to individuals as early as high school and during undergraduate programs [62,79–81]. A higher completed educational level is positively related to the probability of answering an inflation-related question correctly [82]. Similarly, a significant difference of 20 percentage points in FK was found between people who attended college and those who only graduated high school [75]. Educational level influences FL, possibly through its association with other factors, for example, math skills, age, and income [69]. This is confirmed by researchers who have found a link between saving behaviors and the educational levels of subjects [83], as well as a more positive FA for more educated individuals [84]. For example, people with higher educational attainment probably have more income which increases the incentive to correctly manage their finances.

In terms of occupation, it has been shown that individuals who are working show a higher level of FL than those who do not work [73], which may be a result of FEPs at the workplace or experience and skills collected at the job, as knowledge accumulates over time [85]. Regarding entrepreneurs, several studies point to a low level of FL among microentrepreneurs [86]. Similarly, researchers have found that the higher the gross profit ratio of the enterprise, the better the FL scores obtained by entrepreneurs, concluding that microentrepreneurs score worse than those of bigger firms [87]. In Thailand, among a sample of 18 small restaurants, it was found that about 75% of microentrepreneurs show low levels of FK and risk business failure [88]. However, it has been shown that self-employed individuals have better FL scores, compared with those in traditional employment [89,90].

Regarding marital status, several studies have investigated its relationship to FL levels. Those who are married are more likely to be financially literate than those who are single, since married people are more likely to have experience with financial decision-making [91]. Similarly, it was determined that single women were significantly less likely to answer the FL test favorably, than those with a partner [74].

As for the area of residence and its relation to FL, it has been found that urban area residents present higher FL levels as shown in the Russian case study about retirement planning [92]. Rural areas may show lower levels of FL, due to decreased access to resources such as technology [93], and lower educational and income levels [74]. Similar results were encountered in the United States, where they found FL level differences according to geographic location [94]. The Mexican study regarding FL and the use of credit cards showed a difference in the FL score for different regions of the country [40]. Germany [74] and Italy present similar results [95].

Based on the above studies, the three following state hypotheses will help to contrast our results with the conceptual framework:

H1. *Participants present an FL level above the one presented by the OECD.*

H2. *The average FL, FK, FB, and FA scores of participants significantly differ between groups.*

H3. The average FL scores of the participants were significantly different between the categories of five sociodemographic factors (Gender, Age, Marital status, Educational level, and Occupation).

3. Research Design

3.1. Sample

The research is based on one of the Ecuadorian financial institution programs collaborating for this research purpose. The FI's goal was to provide training for approximately 1800 participants classified into three groups: Youth (less than 17 years old), Young Adults (18–24 years old), and Adults (25–65 years old). These segments were considered as a reference taken from a resolution set by the SB [27]. The participants were categorized into 4 subgroups: High-school students (HS), Company employees -public/private- (CE), General public (GP), and financial institution employees (FIE). Although a total of 1775 people from the 4 subgroups were registered, some did not show up and did not comply with the FEP. Only 1723 people received the training through one of two modalities: in situ and online.

From the 1723 trained people, a representative sample of 314 people was obtained using the stratified random sampling technique with 95% confidence [96]. This technique used the classification of the 4 groups of the population previously defined: HS, n = 117; CE, n = 96; GP, n = 54; and FIE, n = 47. Women had greater participation at 51%; students between 15 and 20 years old were the most numerous strata to participate at 37%. A total of 43% of the 314-person sample represents clients of the FI.

The study was conducted in the Guayas province of Ecuador. The 314 participants were from four cantons: Guayaquil (n = 305), Durán (n = 5), Daule (n = 2), and Milagro (n = 2). It should be noted that all participants come from the urban areas.

3.2. Data Collection

First, it is necessary to recognize that the FI's FEP execution is in the in-house format. Mainly, the FI's FEP participants are the FI's clients (individuals or institutions), for example, schools, companies, and the general public are participants of these FEPs. Nevertheless, the approach for institutional participants requires a formal procedure accomplishment. As an example, for educational institutions (schools, high Schools, etc.), the FI offer and send a request to the leading authorities of the educational institutions with the FEP's contents and schedule, to offer the FEP training to their community. The acceptance to participate in the FI's FEP requires the educational facilities and logistic considerations agreements.

For the general public, the press, radio stations, and social networks serve as channels to invite participants to be part of the FI's FEP. The general public interested in the FI's FEP must be enrolled at the FI's registration points.

No previous financial background is needed to join the FI's FEP; the requirement is to be interested in learning.

The FI's FEP is delivered in 2 different modalities. HS and CE groups attended the training in the 'on-site' modality inside the banks' premises, whereas the GP and FIE received the training online through a virtual platform. However, some participants from the GP group also chose the workshop on-site.

The FI's FEP consists of a 2-h educational session, and the proposed scale has been applied as post-evaluation to offer the validity and reliability test. Whether on-site or online, the participants fulfill the evaluation at the end of the last session.

3.3. Informed Consent

All participants of the workshop/training were aware that an evaluation is part of the FI's FEP training experience. This study was aligned with the Declaration of Helsinki and received approval from the institutional authorities of ESPOL Polytechnic University [97]. At the beginning of the session, The participants received a brief statement explaining that the information provided would be used solely for academic purposes, maintaining maximum confidentiality. By remaining in the place for the training, and by signing the form, they confirmed their participation agreement (informed consent) as voluntary and anonymous. They also received the senior researcher's contact information for any current or further support concerning this study.

3.4. Scale Design

The scale design was based on the financial dimensions, and the FEP content related to the ten main subjects established by the SB for all Ecuadorian FIs. These topics were categorized on two different axes.

The axis for all Controlled Systems includes:

- Structure, concepts, actors, and operation of controlled systems;
- Financial planning, saving, and preparation of a family budget;
- Rights and obligations contained in the "Code of Rights of the User of the Financial System";
- Role of the Superintendency of Banks;
- Administration, associated risks, rights, and obligations of credit operations;
- Forms and legal figures established to exercise their rights and claims both within the controlled entity and in public bodies.

The axis of Public and Private Financial System includes:

- Administration, handling, use, rights and obligations, and associated risks of the products offered by the controlled system, such as savings books, current accounts, and use of checks, term deposits, credits, and credit cards, among others;
- Administration, management, use, rights and obligations; and risks associated with the financial services offered, such as debit cards, ATMs, electronic banking, drafts and transfers, and remittances, among others;
- Insurance related to the products offered by the controlled entities, especially about general information on insurance, the rights and obligations of the insured, the risks covered and insurance exclusions, the insured amounts, and the process, requirements, and deadlines to make claims in the event of the loss, among others;
- Use of transactional channels.

According to Appendix A, the items considered in the scale are from different authors and prestigious organizations. Four items are from G20/OECD INFE Core Competencies Framework on FL for Adults [98], four items are from OECD/INFE Core Competencies Framework on FL for Youth [99], two items are from CAF Cuestionario de medición de capacidades financieras de Argentina [100], one is from OECD/INFE International Survey of Adult FL Competencies [101], one is from CAF Cuestionario de medición de capacidades financieras de Ecuador [102], and one is by author [58]. From a total of twenty items, twelve were adapted (A), one was copied and labeled with (C), and seven items (35%) were classified as new (N).

The items are classified into three known dimensions: FK, FB, and FA. The FK subscale has ten items (50% from the total) and is categorized into "basic" and "intermediate" questions, each with five items.

The basic knowledge topics include the role of regulatory entities, consumer protection, and distribution channels of financial services. The intermediate knowledge topics include items regarding budgeting and the traits, uses, and applications of financial products and services. For each item, there is a correct option among the six alternatives.

For the two dimensions FB and FA, a Likert scale was used as a non-comparative scaling technique [103].

The FB dimension subscale has five items focused on the topic of savings, control, and insurance [58,98,99,101]. This dimension identifies the behavior of participants in the face of everyday and family situations. The participant is evaluated through a 6-optioned Likert scale (Not applicable, 0; never, 1; almost never, 2; sometimes, 3; almost always, 4; and always, 5).

Regarding the subscale of the FA where it considers aspects of financial issues for correct decision making, it also has 5 items related to transactional channels, savings, and control, among others.

The attitude subscale considers 5 items related to transactional channels, savings, and control, among others. [98–100,102]. This dimension was measured through a Likert scale, each item with 6 options (Not applicable, 0; totally disagree, 1; disagree, 2; indifferent, 3; agree, 4; totally agree, 5).

The FL score is between 0 and 100 points, calculated from the simple average of the three escalated FA, FB, and FK dimensions to 0–100 points: 25 (100) points of FA (5 times 5), 25 (100) points of FB (5 times 5): 10 (100) points of FK (10 times 1).

In Table S1 of Supplementary Materials, the survey model includes 5 sociodemographic variables additional to the items considered for the scale proposal and validation: gender, age, marital status, educational level, and occupation. Lastly, the survey contains 3 items to describe banking clients' preferences. This includes: You perform financial transactions most frequently in: (ATMS, Online, Mobile, prefer not to share); where do you prefer to keep/invest your savings? (Banks, Cooperative/Mutual, Stock Exchange, none of the above, prefer not to share); and what source of information do you feel is most influential when you decide to choose a product or service from FI? (Guidance and advice from third parties, information provided by the financial institution, my own previous experience, prefer not to share).

3.5. Analysis Procedure

To validate the scale, Exploratory Factor Analysis (EFA) and Confirmatory Factor Analysis (CFA) have been implemented as methodological processes (See Figure 1). Measurement theory can then be combined with structural theory to fully specify an SEM model [104,105]. The fittest and most effective model is possible with the fit measures, reliability, and convergent validity results [30,34].



Figure 1. Analysis Procedure.

All tests and analyses use the R software [106]. The psych package [107] was used to develop the EFA. The lavaan package [108] and semTools [109] help the CFA models through the Diagonally Weighted Least Squares (DWLS) estimation method with adjustments (WLSMV) [110].

The correlation matrix is an essential step in EFA and CFA to identify deniable items, the correlations are polychoric due to the variable's composition, where negative coefficients are omitted as non-representatives [111].

For EFA, Bartlett's test seeks to determine the representative existence of correlations between the variables [112]. KMO has levels that range between 0 and 1; values that will guarantee the permanence/elimination of items: (below 0.50, unacceptable; 0.50 or above, miserable; 0.60 or above, mediocre; 0.70 or above, middling; 0.80 or above, meritorious; 0.90 or above; marvelous; and 1, perfect) [113]. Since the correlations are of the polychoric type, the most appropriate factor extraction method is to use the Weighted Least Squares (WLS) as it has the advantage of easily handling multiple factors and is asymptotically efficient [114,115]. The parallel analysis allows for determining the approximate number of factors [116]. It is important to apply the appropriate type of rotation to improve the interpretation of factor loadings. The chosen rotation method is oblique since it offers the additional advantage of allowing the estimation of the correlations of the factors [117]. The Promax method is used because it efficiently handles large amounts of data compared with the Oblimin method [118].

Due to the sample size (n = 314), 0.35 is the cut-off value for factor loadings to be representative [104].

Table A1 in Appendix A shows the CFA factor loads for the three subscales.

Concerning the fit measures, the respective cut-off values for relative chi-square/normed, chi-square (χ^2/df) range between 2 and 5; then Goodness of Fit Index (GFI) values ≥ 0.95 ; Comparative Fit Index (CFI) values ≥ 0.95 ; Normed Fit Index (NFI) values ≥ 0.95 ; Tucker–Lewis Index (TLI) values ≥ 0.95 ; Standardized Root Mean Squared Residual (SRMR) values < 0.08; and Root Mean Square Error of Approximation (RMSEA) values < 0.08 [119].

The reliability consists of Cronbach's alpha classified as acceptable, between $0.60 \le \alpha < 0.70$ or above; good, between $0.80 \le \alpha < 0.90$; excellent, 0.90 or above [120]. The omega (ω) values ≥ 0.70 [121].

The convergent validity of each construct uses the average variance extracted (AVE) values ≥ 0.50 [122,123].

Finally, through SEM and based on theory, three hypotheses are proposed to support the relationships between the constructs and FL Banking found in the CFA [105]. See Figure 1.

To perform the descriptive analysis FL and its dimensions are the continuous variables of interest. The categorical independent variables are groups (HS = 1, CE = 2, GP = 3, and FIE = 4); gender (male = 1 and female = 2); age (between 15 and 20 years old = 1, between 21 and 26 years old = 2, between 27 and 32 years old = 3, and between 33 and 50 years old = 4); marital status (single = 1, married = 2, free union = 3, divorced = 4); educational level (high school = 1, higher technological level = 2, third level, grade = 3, fourth level, postgraduate = 4); occupation (housewife/househusband = 1, public employee = 2, private employee = 3, microentrepreneur = 4, does not apply = 5).

The inferential analysis evaluates the three proposed hypotheses by T-Student [124] with a *p*-value ≤ 0.05 . We first test the main assumptions: The normality of the errors uses the Shapiro–Wilk test [125]. The homogeneity of variances uses Levene's test [126]. In addition to Cohen's Test to help determine the magnitudes of the effect sizes in both groups [127]: |d| < 0.20 "insignificant"; |d| < 0.50, "small"; |d| < 0.80 "medium"; otherwise "large". Partial Eta Squared (η_p^2) was used to determine the proportion of the variance attributable to the possible effect [128]: the following classification related to the size effect of the variance: $\eta_p^2 > 0.01$ "small"; $\eta_p^2 > 0.06$, "median"; $\eta_p^2 > 0.14$ "large"; otherwise $\eta_p^2 > 0.50$ "maximum". Figure 1 illustrates the whole process.

4. Results

4.1. EFA

Verifying the FL scale structured by FA, FB, FK, Bartlett's test (χ^2 (190) = 54,792.16; p < 0.000 ***) and KMO (KMO = 0.85) present favorable results. The KMO of each item also exceeds the optimal value of 0.70, except for the item FK9 (FK9 = 0.68).

The factor extraction method used was Weighted Least Squares (WLS). The parallel visualization method shows four factors to fit the FL scale; however, comparing the factorial models based on the structure and fit measures, three factors are most likely the ideal number of factors in this exploratory process. The chosen rotation method by comparing Oblimin and Promax highlights promax with better results is as follows: (χ^2 (133) = 47,428; p < 0.000 ***), (SRMR = 0.097, BIC = 46,664). The total variance explained by the three factors is 79%. These are encouraging results regarding the EFA, which allow the development of the CFA.

4.2. CFA

4.2.1. Correlation Matrix Analysis

The correlation matrix of Figure 2 reveals negative relationships (deniable items) that items FK.1 and FK.9 have with the listed items, and positive and strong correlations (undeniable items) FK.2 and FK.3 (r = 0.96), FB.4 and FB.5 (r = 0.94).



Figure 2. Correlation Matrix.

4.2.2. Fit Measures

Table 1 shows the three tested models with eighteen variables (two omitted in the previous section). The third and selected model fit measures (χ^2 (128) = 1.436,45; p < 0.000 ***), GFI = 0.990, CFI = 0.991, NFI = 0.990) include covariances both between latent variables and between items (FK~~FA, FB~~FA, FK~~FB, FK2~~FK3, and FB4~~FB5).

Trial	Model 1	Model 2	Model 3
			FK~~FA
		FK~~FA	FB~~FA
Applied co-variances	-	FB~~FA	FK~~FB
		FK~~FB	FK2~~FK3
			FB4~~FB5
Chi-square (value)	2.062,77	1.450,66	1.436,45
Chi-square (<i>p</i> -value)	0.000 ***	0.000 ***	0.000 ***
Degrees of freedom	133	130	128
Chi-square/Degrees of freedom	15.51	11.16	11.22
GFI-Goodness of fit index	0.984	0.990	0.990
CFI-Comparative fit index	0.984	0.991	0.991
NFI-Normed fit index	0.984	0.990	0.990
TLI-Tucker–Lewis's index	0.982	0.989	0.989
SRMR- Standardized Root mean square residual	0.294	0.279	0.278
RMSEA-Root mean square error of approximation	0.313	0.244	0.244

Table 1. Summary of fit measures for the models.

*** p < 0.001

4.2.3. Measure of Reliability and Validity of Convergence (Convergent Validity)

Table 2 shows the results of the main measures of the quality of the 18-item scale. FL ($\alpha = 0.908$) and FB ($\alpha = 0.956$) are excellent, while FK ($\alpha = 0.794$) and FA ($\alpha = 0.630$) are acceptable. FL, FK, FB, and FA exceed the minimum acceptable value of 0.70 in the McDonald's omega first-order reliability indicator.

Table 2. Measures of Reliability and Convergent Validity.

Subscale	Items	α	ω	AVE
FL	18	0.908	0.962	0.745
FK	8	0.794	0.894	0.706
FB	5	0.956	0.962	0.946
FA	5	0.630	0.808	0.605

Notes: α: Cronbach's alpha; ω: McDonald's omega; AVE: Average variance extracted.

The AVE confirms the robustness of these results, with each construct explaining more than 50% of the indicator variable's variance.

4.2.4. Structural Relationships and Hypothesis Testing

The final model remarks the FK construct with the highest coefficient (43%) among the FL dimensions, having slight differences between FA (29%) and FB (28%) as secondand third-order places.

$$FL = (0.43 * FK) + (0.28 * FB) + (0.29 * FA)$$

The positive relation between dimensions (FA, FB, and FK) and FL are supported in the testing process revealed in Table 3.

Table 3. Structural	l relationships an	ıd hypotheses tes	sting (n = 314).
---------------------	--------------------	-------------------	---------------------

Hypothesis	Path	Estimate	<i>p</i> -Value	Decision
H1	$\mathrm{FL} \gets \mathrm{FK}$	0.43	0.000 ***	Supported
H2	$FL \leftarrow FB$	0.28	0.000 ***	Supported
H3	$FL \leftarrow FA$	0.29	0.000 ***	Supported

*** *p* < 0.001

4.3. Descriptive Analysis and Correlation Analysis

The HS group shows the lower scores in FL and its dimensions; meanwhile, the FIE group leads the high performance in Table 4.

Table 4. The mean and standard deviation FL and its dimensions by Groups.	Table 4. The me	an and standar	d deviation	۱ FL and its	s dimensions	by Groups.
--	-----------------	----------------	-------------	--------------	--------------	------------

	F	Ľ	FK		I	FB		FA	
	M	SD	M	SD	M	SD	M	SD	
HS	<u>59</u>	0.07	55	0.16	49	0.05	72	0.08	
CE	82	0.13	86	0.26	78	0.20	83	0.10	
GP	83	0.14	89	0.23	78	0.20	82	0.11	
FIE	<u>96</u>	0.02	99	0.03	98	0.02	93	0.03	

Table A2 in Appendix A shows high and positive correlations of FL and its dimensions (FL and FK (r = 0.86), FB (r = 0.88), and FA (r = 0.86); and the sociodemographic variables (age, marital status, and educational level) with the strongest relationship between FB and FL.

Table S2 in Supplementary Materials highlights how the FL scores are higher for women in the GP group, though on average men reach the highest score among these groups. Adults (33–50 years old) are the ones with the best overall performance, except for the FIE group where the youngest participants show a slightly higher score. For all the other groups, the youth have the lowest scores.

4.4. Hypotheses

Hypothesis 1 is confirmed. Participants present an FL level above the one presented by the OECD.

The means of FL and the dimensions' score are consistently tested above the 60% benchmark: FL with 76%; FK = 77%; FB = 70%; and FA = 80% from the total sample. The T-Student tested at 95% for FL (t (313) = 16.11; p < 0.000 ***); FK (t (313) = 3.91; p < 0.000 ***); FB (t (313) = 8.06; p < 0.000 ***); and FA (t (313) = 32.55; p < 0.000 ***).

Hypothesis 2 is confirmed. The average FL, FK, FB, and FA scores of participants significantly differ between groups.

Table A3 in Appendix A shows significant differences between the four groups and the FL and its dimensions (one-way Anova). The Bonferroni multiple comparisons test highlights two of the six comparisons have no significant average differences (CE = GP and GP = FIE). The remarks are for the Knowledge dimension (F (3310) = 81.76; p = 0.000 ***, Cohen's d = 0.44). The average of FIE is the highest among the groups.

Hypothesis 3 is confirmed. The average FL scores of the participants were significantly different between the categories of five sociodemographic factors (Gender, Age, Marital status, Educational level, and Occupation).

Table 5 shows the variables age, marital status, educational level, and occupation with significant differences among their categories. Males show slightly higher performances than females, however, this difference was not representative (t (311) = 0.51, p = 0.31, Cohen's d = 0.06).

The groups 27–32 and 33–50 years-old have higher averages between age sections (F (3310) = 147.45, p = 0.000 ***, $\eta_p^2 = 0.59$). Married and cohabiting individuals had the highest averages in marital status, (F (3310) = 32.80, p = 0.000 ***, $\eta_p^2 = 0.24$). In the educational level, (F (3310) = 169.20, p = 0.000 ***, $\eta_p^2 = 0.62$) the higher average is for the third level. Microentrepreneurs and private sector employees in the occupation section reached the highest performance (F (4309) = 180.67, p = 0.000 ***, $\eta_p^2 = 0.70$).

Source	Type II Sum of Squares	df	F-Statistic	<i>p</i> -Value	${\eta_p}^2$	Post Hoc Comparisons-Bonferroni
Age	5.46	3	147.45	0.000 ***	0.59	15-20 < 21-26 $15-20 < 27-32$ $15-20 < 33-50$ $21-26 = 27-32$ $21-26 < 33-50$ $27-32 = 33-50$
Residuals	3.83	310				
Marital status	2.24	3	32.80	0.000 ***	0.24	Single < Married Single < Free Union Single = Divorced Married = Free Union Married = Divorced Free Union = Divorced
Residuals	7.05	310				
Educational level Residuals	5.76	3 310	169.20	0.000 ***	0.62	High school < Higher technological level High school < Third level, grade High school < Fourth level, postgraduate Higher technological level < Third level, grade Higher technological level = Fourth level, postgraduate Third level, grade > Fourth level, postgraduate
Occupation	6.50	4	180.67	0.000 ***	0.70	Housewife/Househusband = Public Employee Housewife/Househusband < Private Employee Housewife/Househusband < Microentrepreneur Housewife/Househusband = Does not apply Public Employee < Private Employee Public Employee < Microentrepreneur Public Employee = Does not apply Private Employee < Microentrepreneur Private Employee > Does not apply Microentrepreneur > Does not apply
Residuals	2.78	309				where the prenetic > Does not apply

Table 5. One-way ANOVA to FL and Sociodemographic factors.

*** *p* < 0.001

4.5. The Bank's Customer Preferences

The results from the three questions regarding the participants' banking preferences are shown in Table S3 of the Supplementary Materials. Of the participants, 70.06% use e-banking, 21.02% prefer ATMs, and 6.05% selected banking agencies. The results remark their preference to save and invest with banks (78.03%), Cooperatives and Mutual Funds (18.79%), less than 1% (0.64%) in the capital market, including 1.59% with no preferences. In total, 94.27% of participants choose financial products or services based on the financial institutions' information.

5. Discussion

The importance of FL has been vastly regarded in several studies throughout the years since it has a significant impact on the financial well-being of any population across the world [3,69,73]. However, different structures and systems across nations have raised the necessity for FL scales that are tailored to a population's specific traits and economy. In the Ecuadorian case, despite the SB's efforts [27], there are no real evaluation tools to date to measure the results obtained from the mandatory FEPs that are being imparted by the FIs in the country.

The scale design topics were based on the country's regulatory framework which include two main axes: controlled systems and public and private financial systems. The items were mostly adapted from well-known scales vastly used in the scientific literature, along with new items created by the authors. These items correspond to one of the three FL dimensions identified as relevant in the conceptual framework (FK, FB, and FA). Later,

we proved the relationship between FL and its dimensions through the proposed SEM hypothesis [30,129].

The sample of the workshop participants was measured in the three dimensions to obtain and validate the FL scale and finally select a plausible, robust, and reliable model ($\alpha = 0.908$, $\omega = 0.962$, AVE = 0.745). The proposed FL scale shows a Cronbach's alpha of 0.908 indicating excellent reliability and internal consistency of the instrument used. Some scales in the LAC region exhibit good reliability but only consider the knowledge dimension, for example in Mexico (Cronbach's alphas = 0.860) [64], or the study on determinants of personal loan costs (Cronbach's alphas > 0.7) in Chile [66]. Comparing our results with other three-dimensional scales, for example, the gender gap study in 2018 (Cronbach's alphas = 0.6) [58], the Brazilian study in 2016 (Cronbach's alphas > 0.7) [67], and the study on credit card use behavior of young adults (Cronbach's alpha > 0.69) [63], ours seems to show a better internal consistency.

Therefore, the selected FL model for FI in the context of Ecuador consists of three subscales that represent each dimension of FL (FK, FB, and FA). The resulting scale (0.43; 0.28; and 0.29) highlighted the importance of FK. Similarly, the most relevant scales mentioned in the literature considered different weights for each dimension, in most cases focusing more heavily on FK as well [31,33,47].

The results indicate a favorable outcome for the Ecuadorian sample given that the average scores measured by the proposed scale have exceeded the values of the OECD in 2020 [15]. However, it should be noted that this sample had previous preparation due to the training provided by the financial institution which is potentially the reason behind these promising results.

An important observation is that the HS group showed the worst FL results across all considered groups and in all dimensions. This is consistent with the U-Shaped model correlating age and FL [46]. This reveals the need for more didactic programs tailored specifically to these age groups [33,64,65,80,130]. It was verified that the group variable influences FL. As expected, the FIE group performed the best in the FL variable considering their previous knowledge and experience within this field. These better results stem from experience and skills gained through time [85].

Regarding sociodemographic variables, men scored slightly higher than women, how-This contradicts the reviewed this difference was not significant. ever, literature [5,58,67], perhaps because all participants received the FEPs, which could reduce the effect of outside social and cultural factors that differ between genders. The best results for FL were scored by the age groups between 27 and 50 [40,46,73], which agrees with the literature which remarks that middle-aged individuals are the most financially literate due to their life experience and being at an age before cognitive decline [46]. Married and cohabitating individuals present the best FL results among the marital status groups concurring with the available literature [74,91]. This may be explained by the greater likelihood of having experienced key financial decisions [91]. People with the highest educational level (third-level education) proved to be the most literate when compared with lower levels [40,83,84]. According to the literature, these results may stem from the greater math and numeracy skills that are associated with education [69]. Among different occupations, microentrepreneurs and private sector employees scored the best FL results. This confirms that individuals working do show better FL scores than those who do not work (housewife/househusband) [73], and the self-employed do better than those in traditional employment [89,90].

Regarding the participants' banking preferences, approximately 70% use online transactions the most frequently, which is encouraging as it would indicate a high level of technological literacy among this group. It would be useful to further study this relationship. Although 1.59% selected none of the above as their preferred method of keeping/investing money, it is important to consider the sample includes high school students, so these results are not too concerning. However, the less than 1% preference for the stock exchange option should be considered by the regulatory framework, perhaps indicating a need for education to achieve a higher level of literacy regarding capital market development. Seeing as the vast majority (94.27%) of the sample use information provided by the financial institution as their most trusted source, it is reassuring that the financial education they are receiving is coming from this medium.

Despite the relevant research results, this study has a few limitations that the reader should bear in mind. It should be acknowledged that the Likert Scales used for measuring various respondents' answers may be considered subjective. This type of response, particularly the ones regarding FA, may allow participants in this study to answer emotionally and give a favorable answer regarding their attitudes or behaviors towards money.

The current investigation was also limited by the number of participants of the FEP that took part in our post-treatment questionnaire, which means it is only a small sample taken at the investigator's convenience. The sample is 100% from urban areas which have been shown from the extensive research literature to have a higher FL score when compared with rural areas [74,92,95]. Moreover, the conclusions drawn from this study are limited to the financial institution whose information was available and therefore cannot be generalized. The sample would ideally include several FIs, whether public or private, so that the FL results are more robust and can better describe the Ecuadorian population. Additionally, it should be noted that the GP group who decided to approach the institution on their own accord may incur a self-selection bias, which would have affected the results. It is recommended to consider a more numerous and randomized sample in future studies to make proper generalizations about the FL levels of the Ecuadorian population.

This investigation has only examined the validity of a proposed scale tailored to the Ecuadorian context. However, it does not evaluate the effect of the FEP on its participants; before and after treatment data should be retrieved and analyzed to study this effect. Finally, it is important to identify the topics and contents of each financial education program, to assess if the proposed scale is directly applicable to every institution. If it is not, the regulatory perspective should be improved to consider all topics established by the SB.

Nevertheless, the relevance of this study is not overshadowed as it contributes to filling in the gap in the research literature, considering there are no other formal scales to this date that measure the FL levels for the programs being imparted by FIs in Ecuador.

6. Conclusions

Previous studies have shown the relative lack of FL in the country [30,68,69], and even when there are efforts in place to counteract this issue, there are no regulations or evaluations to assess the effects of the FEPs properly. The result is a variety of different approaches by each financial institution. According to regulation, FEPs should encompass the fifteen stated topics, however, not all institutions cover all of them. Similarly, the duration of the programs is not always stated, and target groups, methodologies, and channels differ between institutions. Even when comparing online courses with each other, the type of online learning activities, the materials, resources, and assessments are varied between programs, which is why it is difficult to assess the effectiveness of the programs without a validated scale to do so, and this is the reason we propose one. The differences between programs that complicate their comparability call for a standard evaluation tool to assess if the FEPs are positively impacting their users. Otherwise, measures should be taken to prevent institutions from misusing their financial resources. Moreover, it has been almost a decade since FIs in Ecuador were required to impart FEPs, and to assess if there have been any positive effects on society a scale is necessary to monitor the progress being made regarding this issue. The proposed scale has been validated to serve as a baseline subject to future modifications and tests, given that it is possible that the regulation changes over time.

The authors intend to contribute to the efforts towards the standardization of a scale that can evaluate every FEP imparted by FIs in the country and beyond.

Perhaps the most important implication of the study is the need for further research regarding the effect of the FEP on its participants. Future studies should test if the FEP

interventions are effective, comparing before and after FL results. Furthermore, it should be studied if the program is having a positive long-term impact on its participants and is improving financial well-being.

The validated scale should be used in future studies including rural and urban participants to be a better representation of the Ecuadorian population, as well as participants with different income levels. It is of great importance to know if the FI is reaching rural and lower-income individuals as they probably need it the most [40,92,94].

The regulatory entity should consider a more rigorous revision of the topics dictated by the programs of each institution. Although the regulation states the topics to be included, not every institution includes them all. This is important because the same evaluation tool should be used to make comparisons across different institutions that carry out education programs, which is why it should be encouraged to have the evaluation information publicly available.

Most institutions in the country make a differentiation between programs for different age groups. It is not uncommon to have programs dedicated to children and a younger audience [80,81]. It would be useful to know how these programs are different in the topics taught or the methodologies used. Moreover, more research should be focused on making a proper distinction when interpreting FL scale results from these age groups. A score of 59, which is the average for HS in this study, may not necessarily be a negative result, considering their lack of first-hand experience with interacting with the system, financial products and services, and other topics included in the evaluation.

Similarly, the learning delivery methods used in each FEP should be carefully thought out by the institutions to identify the characteristics that make a program effective. Some may be more effective than others considering the needs and characteristics of the target groups. Online learning, for example, has been proven to have varied outcomes depending on student characteristics such as gender, race, and ability [131]. Therefore, training programs for FL should be developed with specific target groups in mind. Other potential factors include technology access which may be limited to rural areas [93], and technological literacy may be a problem for the elderly [132], which is why more effective alternatives such as traditional in situ courses should be explored in this context [133].

A direct implication for policymakers includes the need for more rigorous control over the FEPs, to make sure they are enhancing participants' financial well-being. Furthermore, the institutions' decision-makers would also benefit from avoiding the waste of financial resources. Lastly, future studies would benefit from an up-to-date diagnosis with more detailed information and characteristics of the FEPs carried out by FIs in the region, to assess areas of improvement and get an overview of the programs.

The present study represents a stepping stone for future investigations to evaluate the effect of the FEPs in Ecuador, and to judge if the efforts of FI are fruitful in delivering positive results.

Supplementary Materials: The following supporting information can be downloaded at: https://www.mdpi.com/article/10.3390/su15020996/s1, Table S1: Survey Model; Table S2: Frequency table of FL according to Sociodemographic by Groups. (N:314); Table S3: Frequency table of FL according to Sociodemographic (N:314).

Author Contributions: Conceptualization, S.M.M.-P. and P.E.; methodology, S.M.M.-P., M.V., P.E. and K.P.-R.; software, K.P.-R.; validation, S.M.M.-P., M.V. and P.E.; formal analysis, S.M.M.-P. and K.P.-R.; investigation, S.M.M.-P. and V.R.; resources, S.M.M.-P.; data curation, S.M.M.-P. and K.P.-R.; writing—original draft preparation, V.R.; writing—review and editing, S.M.M.-P., V.R., M.V. and P.E.; visualization, S.M.M.-P.; supervision, S.M.M.-P. and P.E.; project administration, S.M.M.-P.; funding acquisition, S.M.M.-P. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: The study was conducted following the Declaration of Helsinki and approved by the Ethics Committee of ESPOL. Participants were informed about the survey, its implications, and estimated time. Participants could retire from the research with no

consequences. Furthermore, they were informed that their participation was entirely voluntary, and data is confidential and only for research purposes.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: Not applicable.

Acknowledgments: Thanks to the financial institution and the participants for their collaboration. Special thanks to Ana Cristina Aguilar for her support from the Financial master's degree program of ESPOL.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Table A1. The FL Scale and Item Characteristics: Factor Loading. (n:314).

<u> </u>	v.	Factor	Item S	ource
Code	Items	Loading	Author	Action
	FK (correct answer). α 0.794			
FK-1	What type of income do you think should be used for budgeting? [Money owed to us by other	х	2	٨
ГК-1	people]; [Inheritance to receive in the future]; [<u>Fixed income]</u> ; [Variable income]; [None of the above]; [I don't know]	λ	2	А
	What is the regulation in charge of protecting the rights of the Financial User? [Comprehensive			
FK-2	Organic Criminal Code]; [Internal Tax Regime Law]; [Companies Law]; [User Rights Code]; [None	0.83	1	Ν
	of the above]; [I don't know]			
	Regulatory entity of the State that is in charge of regulating and controlling the financial activity of			
FK-3	cooperatives at the national level. [Superintendency of Banks of Ecuador];	0.85	1	Ν
1100	[Superintendence of Popular and Solidarity Economy]; [Superintendency of Companies];	0.00	1	1
	[Superintendency of market power control]; [None of the above]; [I don 't know]			
FK-4	Is the Pledge Credit a ? [Education Credit]; [Credit to buy clothes];	0.90	1	Ν
	[Credit to acquire a movable property]; [Credit to pay debts]; [None of the above]; [I don 't know]			
	Regulatory entity of the State that controls and monitors the financial activity in the country of			
FK-5	banks, mutuals, and financial companies. [Superintendency of Companies];	0.69	1	Ν
	[Superintendency of Banks of Ecuador]; [Superintendence of Popular and Solidarity Economy];			
	[Superintendency of market power control]; [None of the above]; [I don 't know]			
FK-6	What are the financial services distribution channels? [Virtual Parking (Mahila / Talanhang / Transactionally (ATMG); [Agan size and Counterply (Call conterply	0.95	1	Ν
ГК-0	Banking/Mobile/Telephone/Transactional]; [ATMS]; [Agencies and Counters]; [Call center]; [<u>All of the above]</u> ; [I don ´t know]	0.93	1	IN
	What is defined as a demand deposit contract between the bank and the client, which allows one to			
FK-7	earn interest? [Current account]; [Savings account]; [Policy]; [Insurance]; [None of the above]; [I	0.84	1	Ν
1 K-7	don't know]	0.04	1	1 N
	In what kinds of situations can insurance cover us? [Loss of household assets due to theft];			
FK-8	[Unplanned expenses on health issues]; [Loss of income for our family, in the event of death];	0.58	2	А
	[All of the above]; [None of the above]; [I don 't know]		_	
	Remittances are? [Money transfers made by private companies in the same country]; [Money			
EIC O	transfers made by people in the same country];	Ň		
FK-9	[Money transfers made by people from one country to another]; [Money transfers made by public	Х	4	А
	companies in the same country]; [None of the above]; [I don 't know]			
	Who is the customer advocate? [It is a support for clients in the face of any problem related to			
FK-10	financial products and/or services]; [It is a mediator between the user and the Financial Institution];	0.43	1	Ν
1 K-10	[It is a protector of the particular rights and interests of the clients of a financial institution];	0.45	1	1 N
	[<u>All of the above</u>]; [None of the above]; [I don 't know]			
	FB (1 Never–5 Always). α 0.956		_	
FB-1	When you apply to a credit do you know what INSURANCE you are paying?	0.94	5	A
FB-2	Do you create a monthly budget for personal expenses.	0.95	3	A
FB-3	Do you choose products in an informed way.	0.96	3	A
FB-4 FB-5	Do you prefer to buy assets with your own savings rather than to go into debt.	0.89	2 7	A
гр-Э	Do you frequently save at least a minimum percentage of your income. FA (1 Strongly disagree- 5 Completely agree). α 0.630	0.94	1	А
FA-1	Do you consider that you have extensive knowledge of the products offered by FI?	0.81	3	А
FA-2	Do you consider that saving is a positive exercise for your financial growth?	0.64	3	A

Table A1. Cont.

0.1	The second s	Factor	Item Source	
Code	Items	Loading	Author	Action
FA-3	Do you consider that keeping a record of income, expenses, and debts is important?	0.72	2	А
FA-4	Do you rely on transactional channels to enter passwords and inquire about financial products/services?	0.77	4	А
FA-5	Are you are willing to risk some of your own money when making an investment?	0.79	6	С
X: Delete	ach ´s alpha (Cronbach 1951) ed Items after Correlational analysis. e based on seven formal studies.			

The authors, 7 or 35%1.

(G20/OECD INFE Core Competencies Framework on FL for Adults 2016), 4 or 20% 2.

3. (OECD/INFE Core Competencies Framework on FL for Youth 2015), 4 or 20%

(CAF Cuestionario de medición de capacidades financieras de Argentina 2017), 2 or 10% 4.

5. (OECD/INFE International Survey of Adult FL Competencies 2016), 1 or 5%

(CAF Cuestionario de medición de capacidades financieras de Ecuador 2013), 1 or 5%

6. 7. (Potrich, Vieira, and Kirch 2018), 1 or 5%

Twenty Items: 60% Adapted (A), 5% Copy (C), and 35% New (N).

Table A2. Correlations Matrix between FL and Its Dimensions and Sociodemographic Factors.

	FL	FK	FB	FA	Gender	Age	Marital Status	Educational Level	Occupation
FL	1.00								
FK	0.86 ***	1.00							
FB	0.88 ***	0.54 ***	1.00						
FA	0.86 ***	0.58 ***	0.81 ***	1.00					
Gender	-0.04	-0.11	0.04	0.00	1.00				
Age	0.72	0.54	0.76	0.62	0.08	1.00			
Marital status	0.64	0.43	0.66	0.63	0.10	0.88	1.00		
Educational level	0.72	0.54	0.76	0.58	0.06	0.77	0.43	1.00	
Occupation	-0.49	-0.41	-0.52	-0.33	-0.09	-0.61	-0.19	-0.80	1.00

*** *p* < 0.001(Pearson). Notes: Pearson FL, FK, FB, FA; Polyserial FL, FK, FB, FA and sociodemographic factors; Polychoric sociodemographic factors.

S	Source	Type II Sum of Squares	df	F-Statistic	<i>p</i> -Value	η_p^2	Post Hoc Comparisons-Bonferroni
							HS < CE
							HS < GP
FL	Crowns	5.99	3	188.12	0.000 ***	0.64	HS < FIE
FL	Groups	5.99	3	188.12	0.000	0.64	CE = GP
							CE < FIE
							GP < FIE
	Residuals	3.29	310				
							HS < CE
							HS < GP
FK	Groups	9.45	3	81.76	0.000 ***	0.44	HS < FIE
110	Groups	2.10	0	01.70	0.000	0.11	CE = GP
							CE < FIE
							GP = FIE
	Residuals	11.94	310				
							HS < CE
							HS < GP
FB	Groups	9.83	3	163.57	0.000 ***	0.61	HS < FIE
	1						CE = GP
							CE < FIE GP < FIE
	Residuals	6.21	310				GP < FIE
	Residuals	0.21	510				

Table A3. One-way ANOVA to FL and its dimensions by groups.

S	ource	Type II Sum of Squares	df	F-Statistic	<i>p</i> -Value	η_p^2	Post Hoc Comparisons-Bonferroni
							HS < CE
							HS < GP
FA	Crowns	1.52	3	68.37	0.000 ***	0.40	HS < FIE
гA	Groups	1.32	3	00.37	0.000	0.40	CE = GP
							CE < FIE
							GP < FIE
	Residuals	2.29	310				
		*** <i>p</i> < 0.001					

Table A3. Cont.

References

- Tavares, F.; Santos, E.; Tavares, V.; Ratten, V. The Perception and Knowledge of Financial Risk of the Portuguese. *Sustainability* 2020, 12, 8255. [CrossRef]
- Mitchell, O.S.; Schieber, S.J. Living with Defined Contribution Pensions: Remaking Responsibility for Retirement; Pension Research Council, Wharton School of the University of Pennsylvania: Philadelphia, PA, USA, 1998.
- Lusardi, A.; Mitchell, O. Financial Literacy and Planning: Implications for Retirement Wellbeing; National Bureau of Economic Research: Cambridge, MA, USA, 2011. Available online: http://www.nber.org/papers/w17078.pdf (accessed on 16 November 2022).
- Lusardi, A.; Mitchelli, O.S. Financial Literacy and Retirement Preparedness: Evidence and Implications for Financial Education. Bus. Econ. 2007, 42, 35–44. [CrossRef]
- 5. Lusardi, A.; Mitchell, O. Planning and Financial Literacy: How Do Women Fare? Am. Econ. Rev. 2008, 98, 413–417. [CrossRef]
- Anderson, A.; Baker, F.; Robinson, D.T. Precautionary Savings, Retirement Planning and Misperceptions of Financial Literacy. J. Financ. Econ. 2017, 126, 383–398. [CrossRef]
- 7. van Nguyen, H.; Ha, G.H.; Nguyen, D.N.; Doan, A.H.; Phan, H.T. Understanding Financial Literacy and Associated Factors among Adult Population in a Low-Middle Income Country. *Heliyon* **2022**, *8*, e09638. [CrossRef]
- Grohmann, A. Financial Literacy and Financial Behavior: Evidence from the Emerging Asian Middle Class. *Pac. Basin Financ. J.* 2018, 48, 129–143. [CrossRef]
- 9. Boisclair, D.; Lusardi, A.; Michaud, P.C. Financial Literacy and Retirement Planning in Canada. J. Pension Econ. Financ. 2017, 16, 277–296. [CrossRef]
- 10. van Rooij, M.; Lusardi, A.; Alessie, R. Financial Literacy and Stock Market Participation. J. Financ. Econ. 2011, 101, 449–472. [CrossRef]
- 11. Mouna, A.; Anis, J. Financial Literacy in Tunisia: Its Determinants and Its Implications on Investment Behavior. *Res. Int. Bus. Financ.* 2017, *39*, 568–577. [CrossRef]
- 12. Hassan Al-Tamimi, H.A.; Anood Bin Kalli, A. Financial Literacy and Investment Decisions of UAE Investors. J. Risk Financ. 2009, 10, 500–516. [CrossRef]
- 13. Merkoulova, Y.; Veld, C. Why Do Individuals Not Participate in the Stock Market? *Int. Rev. Financ. Anal.* 2022, *83*, 102292. [CrossRef]
- 14. CFPB. Financial Literacy Annual Report; CFPB: Washington, DC, USA, 2016.
- 15. OECD. OECD/INFE 2020 International Survey of Adult Financial Literacy; OECD: Paris, France, 2020. Available online: https://www.oecd.org/financial/education/launchoftheoecdinfeglobalfinancialliteracysurveyreport.htm (accessed on 26 November 2022).
- 16. OECD Council. Recommendation on Principles and Good Practices for Financial Education and Awareness; OECD: Paris, France, 2005.
- 17. Méndez Prado, S.M.; Zambrano Franco, M.J.; Zambrano Zapata, S.G.; Chiluiza García, K.M.; Everaert, P.; Valcke, M. A Systematic Review of Financial Literacy Research in Latin America and The Caribbean. *Sustainability* **2022**, *14*, 3814. [CrossRef]
- 18. Silva, T.C.; Braz, T.; Amancio, D.R.; Tabak, B.M. Financial Literacy and the Perceived Value of Stress Testing: An Experiment Using Students in Brazil. *Emerg. Mark. Financ. Trade* 2022, *58*, 965–996. [CrossRef]
- 19. Alí, Í.; Sepúlveda, J.; Sepúlveda, J.; Denegri, M. The Impact of Attitudes on Behavioural Change: A Multilevel Analysis of Predictors of Changes in Consumer Behaviour. *Rev. Lat. Psicol.* **2021**, *53*, 73–82. [CrossRef]
- Paraboni, A.L.; Soares, F.M.; Potrich, A.C.G.; Vieira, K.M. Does Formal and Business Education Expand the Levels of Financial Education? *Int. J. Soc. Econ.* 2020, 47, 769–785. [CrossRef]
- Méndez Prado, S.M.; Everaert, P.; Valcke, M. Is the Dosed Video-Vignettes Intervention More Effective with a Longer-Lasting Effect? A Financial Literacy Study. In Proceedings of the 2020 12th International Conference on Education Technology and Computers, London, UK, 23 October 2020; ACM: New York, NY, USA, 2020; pp. 193–198.
- 22. Paraboni, A.L.; da Costa, N. Improving the Level of Financial Literacy and the Influence of the Cognitive Ability in This Process. *J. Behav. Exp. Econ.* **2021**, *90*, 101656. [CrossRef]

- Comité de Educación Financiera Estrategia Nacional de Educación FInanciera México. Available online: https://www.gob.mx/ forodeinclusionfinanciera/articulos/estrategia-nacional-de-educacion-financiera-enef?idiom=es#:~{}:text=La%20Estrategia% 20Nacional%20de%20Educaci%C3%B3n,bienestar%20financiero%20de%20la%20poblaci%C3%B3n (accessed on 26 September 2022).
- En Brasil—ENEF. Available online: https://www.vidaedinheiro.gov.br/es/educacao-financeira-no-brasil/?doing_wp_cron=16 60939647.5076899528503417968750 (accessed on 18 August 2022).
- 25. *Estrategia Nacional de Educación Financiera Chile*; Gobierno de Chile: Santiago, Chile, 2017. Available online: https://bibliotecadigital.mineduc.cl/handle/20.500.12365/2167 (accessed on 26 November 2022).
- 26. *Experiencias Y Aprendizajes De La Educación Financiera;* Asobancaria: Bogotá, Colombia, 2016. Available online: https://www.sabermassermas.com/experiencias-y-aprendizajes-de-la-educacion-financiera/ (accessed on 19 August 2022).
- Superintendencia de Bancos del Ecuador. *Resolución SB-2015-665*; Superintendencia de Bancos del Ecuador: Quito, Ecuador, 2015. Available online: https://www.superbancos.gob.ec/bancos/wp-content/uploads/downloads/2018/04/resol_SB-2015-665.pdf (accessed on 27 September 2022).
- Quiénes Somos-Finanzas Personales Ecuador. Available online: https://tusfinanzas.ec/quienes-somos/ (accessed on 25 September 2022).
- 29. Méndez Prado, M.; Marcillo, Q.L. La Educación Financiera Toma Rumbo En El País | Gestión. 2018. Available online : https://revistagestion.ec/economia-y-finanzas-analisis/la-educacion-financiera-toma-rumbo-en-el-pais (accessed on 12 September 2022).
- Méndez Prado, S.M.; Chiluiza, K.; Everaert, P.; Valcke, M. Design and Evaluation among Young Adults of a Financial Literacy Scale Focused on Key Financial Decisions. *Educ. Sci.* 2022, 12, 460. [CrossRef]
- 31. OECD. OECD/INFE Toolkit for Measuring Financial Literacy and Financial Inclusion; OECD: Paris, France, 2018.
- 32. Klapper, L.; Lusardi, A.; van Oudheusden, P. Financial Literacy Around the World: Insights from the S&P Global Financial Literacy Survey; GFLEC: Washington, DC, USA, 2015.
- Walstad, W.B.; Rebeck, K. The Test of Financial Literacy: Development and Measurement Characteristics. J. Econ. Educ. 2017, 48, 113–122. [CrossRef]
- 34. Schuhen, M.; Schürkmann, S. Construct Validity of Financial Literacy. Int. Rev. Econ. Educ. 2014, 16, 1–11. [CrossRef]
- 35. Huston, S.J. Measuring Financial Literacy. J. Consum. Aff. 2010, 44, 296–316. [CrossRef]
- Lusardi, A.; Mitchell, O.S. The Economic Importance of Financial Literacy: Theory and Evidence. J. Econ. Lit. 2014, 52, 5–44. [CrossRef] [PubMed]
- 37. OECD. PISA 2012 Results: Students and Money; PISA; OECD: Paris, France, 2014; Volume 5, ISBN 9789264208087.
- Atkinson, A.; Messy, F.-A. Measuring Financial Literacy: Results of the OECD/International Network on Financial Education (INFE) Pilot Study. In OECD Working Papers on Finance, Insurance and Private Pensions; OECD: Paris, France, 2012; Volume 15. [CrossRef]
- Muñoz-Murillo, M.; Álvarez-Franco, P.B.; Restrepo-Tobón, D.A. The Role of Cognitive Abilities on Financial Literacy: New Experimental Evidence. J. Behav. Exp. Econ. 2020, 84, 101482. [CrossRef]
- 40. Hernández-Mejía, S.; García-Santillán, A.; Moreno-García, E. Financial Literacy and the Use of Credit Cards in Mexico. *J. Int. Stud.* **2021**, *14*, 97–112. [CrossRef]
- Duch, R.; Granados, P.; Laroze, D.; López, M.; Ormeño, M.; Quintanilla, X. Choice Architecture Improves Pension Selection. *Appl. Econ.* 2021, 53, 2256–2274. [CrossRef]
- 42. Santoyo Ledesma, D.S.; Luna Nemecio, J. Experiencia Exploratoria de Validación de Un Instrumento Sobre Nivel de Cultura Financiera En La Generación Millennial. *Rev. de Métodos Cuantitativos Para La Econ. y La Empresa* **2021**, *31*, 226–239. [CrossRef]
- 43. Financial Literacy and Inclusion Results of OECD/INFE Survey across Countries and by Gender; OECD: Paris, France, 2013.
- 44. Allianz. When Will the Penny Drop? Allianz: Munich, Germany, 2017.
- 45. Ouachani, S.; Belhassine, O.; Kammoun, A. Measuring Financial Literacy: A Literature Review. *Manag. Financ.* 2020, 47, 266–281. [CrossRef]
- Agarwal, S.; Driscoll, J.C.; Gabaix, X.; Laibson, D. The Age of Reason: Financial Decisions over the Life Cycle and Implications for Regulation. *Brook. Pap. Econ. Act.* 2009, 2009, 51–117. [CrossRef]
- 47. Klapper, L.F.; Lusardi, A.; Panos, G.A.; Thank, W.; Al-Hussainy, E.; Markov, A.; Mckenzie, D.; Melecky, M.; Rutledge, S.; Zia, B. *Financial Literacy and the Financial Crisis*; National Bureau of Economic Research: Washington, DC, USA, 2012.
- 48. Christiansen, C.; Joensen, J.S.; Rangvid, J. Are Economists More Likely to Hold Stocks? Rev. Financ. 2008, 12, 465–496. [CrossRef]
- 49. Lusardi, A.; Mitchell, O. *How Ordinary Consumers Make Complex Economic Decisions: Financial Literacy and Retirement Readiness;* National Bureau of Economic Research: Cambridge, MA, USA, 2009.
- 50. Behrman, J.R.; Mitchell, O.S.; Soo, C.K.; Bravo, D. How Financial Literacy Affects Household Wealth Accumulation. *Am. Econ. Rev.* 2012, 102, 300–304. [CrossRef]
- Shockey, S.S. Low-Wealth Adults' Financial Literacy, Money Management Behaviors, and Associated Factors, Including Critical Thinking; The Ohio State University: Columbus, OH, USA, 2002.
- 52. Vieira, K.M.; Potrich, A.C.G.; Mendes-Da-Silva, W. A Financial Literacy Model for University Students. In *Individual Behaviors and Technologies for Financial Innovations*; Springer: Cham, Switzerland, 2019; pp. 69–95. [CrossRef]

- 53. Chu, Z.; Wang, Z.; Xiao, J.J.; Zhang, W. Financial Literacy, Portfolio Choice and Financial Well-Being. Soc. Indic. Res. 2017, 132, 799–820. [CrossRef]
- 54. Serido, J.; Shim, S.; Mishra, A.; Tang, C. Financial Parenting, Financial Coping Behaviors, and Well-Being of Emerging Adults. *Fam. Relat.* **2010**, *59*, 453–464. [CrossRef]
- 55. Philippas, N.D.; Avdoulas, C. Financial Literacy and Financial Well-Being among Generation-Z University Students: Evidence from Greece. *Eur. J. Financ.* 2020, *26*, 360–381. [CrossRef]
- 56. Pandey, A.; Ashta, A.; Spiegelman, E.; Sutan, A. Catch Them Young: Impact of Financial Socialization, Financial Literacy and Attitude towards Money on Financial Well-Being of Young Adults. *Int. J. Consum. Stud.* **2020**, *44*, 531–541. [CrossRef]
- Limbu, Y.B.; Sato, S. Credit Card Literacy and Financial Well-Being of College Students: A Moderated Mediation Model of Self-Efficacy and Credit Card Number. *Int. J. Bank Mark.* 2019, 37, 991–1003. [CrossRef]
- 58. Potrich, A.C.G.; Vieira, K.M.; Kirch, G. How Well Do Women Do When It Comes to Financial Literacy? Proposition of an Indicator and Analysis of Gender Differences. *J. Behav. Exp. Financ.* **2018**, *17*, 28–41. [CrossRef]
- Dogra, P.; Kaushal, A.; Sharma, R.R. Antecedents of the Youngster's Awareness About Financial Literacy: A Structure Equation Modelling Approach. Vis. J. Bus. Perspect. 2021, 0, 097226292199656. [CrossRef]
- 60. Walstad, W.B.; Rebeck, K. Test of Financial Literacy: Examiner's Manual; Council for Economic Education: New York, NY, USA, 2016.
- 61. Förster, M.; Happ, R.; Molerov, D. Using the U.S. Test of Financial Literacy in Germany—Adaptation and Validation. *J. Econ. Educ.* **2017**, *48*, 123–135. [CrossRef]
- 62. Amagir, A.; Groot, W.; van den Brink, H.M.; Wilschut, A. Financial Literacy of High School Students in the Netherlands: Knowledge, Attitudes, Self-Efficacy, and Behavior. *Int. Rev. Econ. Educ.* **2020**, *34*, 100185. [CrossRef]
- 63. Santos, D.B.; Mendes-Da-Silva, W.; Norvilitis, J.M.; Protin, P.; Onusic, L. Parents Influence Responsible Credit Use in Young Adults: Empirical Evidence from the United States, France, and Brazil. *J. Fam. Econ. Issues* **2022**, *43*, 368–383. [CrossRef]
- 64. Garcia-Santillan, A. Knowledge and Application toward Financial Topics in High School Students: A Parametric Study. *Eur. J. Educ. Res.* **2020**, *9*, 905–919. [CrossRef]
- 65. Moreno-García, E.; García-Santillán, A.; Gutiérrez, A.D. los S. Financial Literacy of "Telebachillerato" Students: A Study of Perception, Usefulness and Application of Financial Tools. *Int. J. Educ. Pract.* **2019**, *7*, 168–183. [CrossRef]
- 66. Farías, P. Determinants of Knowledge of Personal Loans' Total Costs: How Price Consciousness, Financial Literacy, Purchase Recency and Frequency Work Together. *J. Bus. Res.* 2019, 102, 212–219. [CrossRef]
- 67. Campara, J.P.; Mendes Vieira, K. "Bolsa Família X" Program Financial Literacy: In Search of a Model for Low-Income Women. *Espacios* **2016**, *37*, 22.
- Arbache, J.; Mejía, D.; Andrés, P.; García, F.; Carrera Vaca, C.; Guevara, G.E.; Rojas, M.J. Encuesta De Medición De Capacidades Financieras Ecuador 2020. 2020. Available online: https://www.findevgateway.org/es/publicacion/2022/07/encuesta-demedicion-de-capacidades-financieras-ecuador-2020 (accessed on 27 September 2022).
- 69. Klapper, L.; Lusardi, A. Financial Literacy and Financial Resilience: Evidence from around the World. *Financ. Manag.* **2020**, *49*, 589–614. [CrossRef]
- 70. Citybank. Programa de Educación Financiera PEF 2022-Banco de Desarrollo; Citybank N.A. Sucursal Ecuador: Quito, Ecuador, 2022.
- 71. CNMV. Plan de Educación Financiera; Banco Central Del Ecuador; CNMV: Madrid, Spain, 2020.
- 72. Citybank. Programa de Educación Financiera PEF 2021; Citybank N.A. Sucursal Ecuador: Quito, Ecuador, 2021.
- 73. Lusardi, A.; Mitchell, O.S. Financial Literacy around the World: An Overview. *J. Pension Econ. Financ.* 2011, 10, 497–508. [CrossRef] [PubMed]
- 74. Bucher-Koenen, T.; Lusardi, A. *Financial Literacy and Retirement Planning in Germany*; National Bureau of Economic Research: Cambridge, MA, USA, 2011.
- Lusardi, A.; Mitchell, O.; Curto, V. Financial Literacy among the Young: Evidence and Implications for Consumer Policy; National Bureau of Economic Research: Cambridge, MA, USA, 2009. Available online: http://www.nber.org/papers/w15352.pdf (accessed on 16 November 2022).
- 76. Swiecka, B.; Yeşildağ, E.; Özen, E.; Grima, S. Financial Literacy: The Case of Poland. Sustainability 2020, 12, 700. [CrossRef]
- 77. Calamato, M.P. Learning Financial Literacy in the Family; San Jose State University: San Jose, CA, USA, 2010.
- 78. Liaqat, F.; Mahmood, K.; Ali, F.H. Demographic and Socio-Economic Differences in Financial Information Literacy among University Students. *Inf. Dev.* **2021**, *37*, 376–388. [CrossRef]
- 79. Gerrans, P.; Heaney, R. The Impact of Undergraduate Personal Finance Education on Individual Financial Literacy, Attitudes and Intentions. *Account. Financ.* 2019, 59, 177–217. [CrossRef]
- Urban, C.; Schmeiser, M.; Collins, J.M.; Brown, A. The Effects of High School Personal Financial Education Policies on Financial Behavior. *Econ. Educ. Rev.* 2020, 78, 101786. [CrossRef]
- 81. Bruhn, M.; de Souza Leão, L.; Legovini, A.; Marchetti, R.; Zia, B. The Impact of High School Financial Education: Evidence from a Large-Scale Evaluation in Brazil. *Am. Econ. J. Appl. Econ.* **2016**, *8*, 256–295. [CrossRef]
- Hernández Mejía, S.; García-Santillán, A.; Moreno-García, E. Financial Literacy and Its Relationship with Sociodemographic Variables. *Econ. Sociol.* 2022, 15, 40–55. [CrossRef]
- 83. Agarwal, S.; Chomsisengphet, S.; ZHANG, Y. How Does Financial Literacy Affect Mortgage Default? *SSRN Electron. J.* **2015**. [CrossRef]

- Azlan, A.; Jamal, A.; Kamal, W.; Mohdrahimie, R.; Roslemohidin, A.K.; Osman, Z. The Effects of Social Influence and Financial Literacy on Savings Behavior: A Study on Students of Higher Learning Institutions in Kota Kinabalu, Sabah. *Int. J. Bus. Soc. Sci.* 2015, *6*, 110–119.
- 85. Kiliyanni, A.L.; Sivaraman, S. The Perception-Reality Gap in Financial Literacy: Evidence from the Most Literate State in India. *Int. Rev. Econ. Educ.* **2016**, *23*, 47–64. [CrossRef]
- Mashizha, M.; Sibanda, M.; Maumbe, B. Financial Literacy among Small and Medium Enterprises in Zimbabwe. S. Afr. J. Entrep. Small Bus. Manag. 2019, 11, 10. [CrossRef]
- Singla, A.; Mallik, G. Determinants of Financial Literacy: Empirical Evidence from Micro and Small Enterprises in India. *Asia Pac. Manag. Rev.* 2021, 26, 248–255. [CrossRef]
- Rachapaettayakom, P.; Wiriyapinit, M.; Cooharojananone, N.; Tanthanongsakkun, S.; Charoenruk, N. The Need for Financial Knowledge Acquisition Tools and Technology by Small Business Entrepreneurs. J. Innov. Entrep. 2020, 9, 25. [CrossRef]
- 89. Struckell, E.M.; Patel, P.C.; Ojha, D.; Oghazi, P. Financial Literacy and Self Employment—The Moderating Effect of Gender and Race. *J. Bus. Res.* 2022, 139, 639–653. [CrossRef]
- 90. Ćumurović, A.; Hyll, W. Financial Literacy and Self-Employment. J. Consum. Aff. 2019, 53, 455–487. [CrossRef]
- 91. Danes, S.M.; Hira, T.K. Money Management Knowledge of College Students. J. Stud. Financ. Aid **1987**, 17, 1. [CrossRef]
- 92. Klapper, L.F.; Panos, G.A. Financial Literacy and Retirement Planning: The Russian Case. SSRN Electron. J. 2011, 10, 599–618. [CrossRef]
- 93. Dow-Fleisner, S.J.; Seaton, C.L.; Li, E.; Plamondon, K.; Oelke, N.; Kurtz, D.; Jones, C.; Currie, L.M.; Pesut, B.; Hasan, K.; et al. Internet Access Is a Necessity: A Latent Class Analysis of COVID-19 Related Challenges and the Role of Technology Use among Rural Community Residents. *BMC Public Health* 2022, 22, 845. [CrossRef]
- 94. Bumcrot, C.; Lin, J.; Lusardi, A. The Geography of Financial Literacy. Numeracy 2013, 6. [CrossRef]
- 95. Fornero, E.; Monticone, C. Financial Literacy and Pension Plan Participation in Italy. J. Pension Econ. Financ. 2011, 10, 547–564. [CrossRef]
- 96. Parsons, V.L. Stratified Sampling. In Wiley StatsRef: Statistics Reference Online; Wiley: Hoboken, NJ, USA, 2017; pp. 1–11.
- 97. Monsalve, C. ESPOL Ethics Committee Certification. 2021. Available online: https://drive.google.com/file/d/1K8dUsrOdo2F2 -YmFRTxj4OmeR8oYTSUX/view?usp=sharing (accessed on 2 January 2023).
- 98. OECD. G20/OECD INFE Core Competencies Framework on Financial Literacy for Adults; OECD: Paris, France, 2016.
- 99. OECD. OECD/INFE Core Competencies Framework on Financial Literacy for Youth; OECD: Paris, France, 2015.
- Iglesias, M.; Mejía, D. Encuesta de Medición de Capacidades Financieras En Argentina; CAF: Buenos Aires, Argentina, 2018. Available online: http://scioteca.caf.com/handle/123456789/1341 (accessed on 27 September 2022).
- OECD. OECD/INFE International Survey of Adult Financial Literacy Competencies; OECD: Paris, France, 2016; Available online: www.oecd.org/finance/OECD-INFE-International-Survey-of-Adult-Financial-Literacy-Competencies.pdf (accessed on 26 November 2022).
- 102. CAF. Cuestionario de Medición de Capacidades Financieras de Ecuador 2013; CAF: Buenos Aires, Argentina, 2013.
- 103. Swaminathan, V. Noncomparative Scaling Technique. In *Wiley International Encyclopedia of Marketing*; John Wiley & Sons, Ltd.: Chichester, UK, 2010.
- 104. Hair, J.F.; Black, W.C.; Babin, B.J.; Anderson, R.E. Multivariate Data Analysis, 7th ed.; Pearson: New York, NY, USA, 2010.
- Ganesan, Y.; Pitchay, A.B.A.; Nasser, M.A.M. Does Intention Influence the Financial Literacy of Depositors of Islamic Banking? A Case of Malaysia. Int. J. Soc. Econ. 2020, 47, 675–690. [CrossRef]
- 106. *R Core Team R: A Language and Environment for Statistical Computing;* R Foundation for Statistical Computing: Vienna, Austria, 2022.
- 107. Revelle, W. (*psych Package*) *psych: Procedures for Personality and Psychological Research*, R package version 2.2.9; Northwestern University: Evanston, IL, USA, 2022.
- 108. Rosseel, Y. lavaan: An R Package for Structural Equation Modeling. J. Stat. Softw. 2012, 48, 1–36. [CrossRef]
- Jorgensen, T.D.; Pornprasertmanit, S.; Schoemann, A.M.; Rosseel, Y. SemTools: Useful Tools for Structural Equation Modeling, R Package Version 0.5-6; 2022. Available online: https://cran.r-project.org/web/packages/semTools/index.html (accessed on 19 August 2022).
- 110. Muthén, B.O. Goodness of Fit with Categorical and Other Non-Normal Variables. In *Testing Structural Equation Models;* Sage: Thousand Oaks, CA, USA, 1993.
- 111. Lorenzo-Seva, U.; Ferrando, P.J. Not Positive Definite Correlation Matrices in Exploratory Item Factor Analysis: Causes, Consequences and a Proposed Solution. *Struct. Equ. Model.* **2021**, *28*, 138–147. [CrossRef]
- 112. Barlett, M.S. The Effect of Standardization on a X2 Approximation in Factor Analysis. Biometrika 1951, 38, 337–344. [CrossRef]
- 113. Kaiser, H.F. An Index of Factorial Simplicity. *Psychometrika* **1974**, *39*, 31–36. [CrossRef]
- 114. Wirth, R.J.; Edwards, M.C. Item Factor Analysis: Current Approaches and Future Directions. *Psychol. Methods* **2007**, *12*, 58–79. [CrossRef]
- 115. Barendse, M.T.; Oort, F.J.; Timmerman, M.E. Using Exploratory Factor Analysis to Determine the Dimensionality of Discrete Responses. *Struct. Equ. Model. A Multidiscip. J.* 2015, 22, 87–101. [CrossRef]
- 116. Horn, J.L. A Rationale and Test for the Number of Factors in Factor Analysis. Psychometrika 1965, 30, 179–185. [CrossRef]
- 117. Preacher, K.J.; MacCallum, R.C. Repairing Tom Swift's Electric Factor Analysis Machine. Underst. Stat. 2003, 2, 13–43. [CrossRef]

- 118. Hendrickson, A.E.; White, P.O. Promax: A Quick Method for Rotation to Oblique Simple Structure. *Br. J. Stat. Psychol.* **1964**, 17, 65–70. [CrossRef]
- 119. Coughlan, J.; Hooper, D.; Mullen, M.R. Structural Equation Modeling: Guidelines for Determining Model Fit. *Electron. J. Bus. Res. Methods* **2007**, *6*, 53–60.
- 120. Cronbach, L.J. Coefficient Alpha and the Internal Structure of Tests. Psychometrika 1951, 16, 297–334. [CrossRef]
- 121. McDonald, R.P. Test Theory; Psychology Press: New York, NY, USA, 2013; ISBN 9781410601087.
- 122. Fornell, C.; Larcker, D.F. Evaluating Structural Equation Models with Unobservable Variables and Measurement Error. *J. Mark. Res.* **1981**, *18*, 39. [CrossRef]
- Hair, J.F.; Hult, G.T.M.; Ringle, C.M.; Sarstedt, M.; Danks, N.P.; Ray, S. Partial Least Squares Structural Equation Modeling (PLS-SEM) Using R; Classroom Companion: Business; Springer: Cham, Switzerland, 2021; ISBN 978-3-030-80518-0.
- 124. William, S. The Probable Error of a Mean. Biometrika 1908, 6, 1. [CrossRef]
- 125. Shapiro, S.S.; Wilk, M.B. An Analysis of Variance Test for Normality (Complete Samples). Biometrika 1965, 52, 591. [CrossRef]
- 126. Keyes, T.K.; Levy, M.S. Analysis of Levene's Test under Design Imbalance. J. Educ. Behav. Stat. 1997, 22, 227. [CrossRef]
- 127. Cohen, J. Statistical Power Analysis for the Behavioral Sciences; Routledge: New York, NY, USA, 2013; ISBN 9781134742707.
- 128. Cohen, J. Eta-Squared and Partial Eta-Squared in Fixed Factor Anova Designs. Educ. Psychol. Meas. 1973, 33, 107–112. [CrossRef]
- Pavkovic, A.; Andelinovic, M.; Misevic, D. Measuring Financial Literacy of University Students. Croat. Oper. Res. Rev. 2018, 9, 87–97. [CrossRef]
- Coria, M.D.; Concha-Salgado, A.; Aravena, J.S. Adaptation and Validation of the Economic and Financial Literacy Test for Chilean Secondary Students. *Rev. Lat. Psicol.* 2019, 51, 113–122. [CrossRef]
- 131. Nguyen, T.D. The Effectiveness of Online Learning: Beyond No Significant Difference and Future Horizons. *MERLOT J. Online Learn. Teach.* **2015**, *11*, 309–319.
- 132. Nguyen, T.X.H.; Tran, T.B.N.; Dao, T.B.; Barysheva, G.; Nguyen, C.T.; Nguyen, A.H.; Lam, T.S. Elderly People's Adaptation to the Evolving Digital Society: A Case Study in Vietnam. *Soc. Sci.* **2022**, *11*, 324. [CrossRef]
- Salehi, A.; Strawderman, L.; Huang, Y.; Ahmed, S.; Babski-Reeves, K. Effectiveness of Three Training Delivery Methods in a Voluntary Program. *Proc. Hum. Factors Ergon. Soc.* 2009, *3*, 1959–1963. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.